
News from CERN

• there was a power cut on Saturday

• comm rack and crates now are back ON

• nPDM GBTx configured, but I did not get 
the standard message “Idle 18h” but 
something like: “WaitDSEL A0 h”  should 
be checked 

• daqux01 not reachable yet.... (this morning 
at 9:30), Loic is informed and is working on 
that (already yesterday afternoon...). if 
nothing happens  I’ll ask again Loic. 

• Federico Alessio and Paolo Durante 
installed another PCIe40 in “our” server, 
will be used by CALO  take that into 
account from now on (multiple  boards in 
the same server) 
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